Author's response to reviews

Title: Hospitalisations during the last year of life of nursing home residents: A retrospective cohort study from Germany

Authors:

Heribert H Ramroth (hramroth@urz.uni-heidelberg.de)
Norbert N Specht-Leible (nspechtleible@bethanien-heidelberg.de)
Hans-Helmut H-H Koenig (hans-helmut.koenig@medizin.uni-leipzig.de)
Hermann H Brenner (brenner@dzfa.uni-heidelberg.de)

Version: 2 Date: 26 October 2005

Author's response to reviews:

Reference citations in the text have been corrected ordered now numerically how they appear in the text. Additionally, the comments from the covering letter have been added to the manuscript.